Author's response to reviews

Title: Relaxation training for anxiety: a ten-years systematic review with meta-analysis

Authors:

Gian Mauro Manzoni (gm.manzoni@auxologico.it)
Francesco Pagnini (francesco.pagnini@unibg.it)
Gianluca Castelnuovo (gianluca.castelnuovo@auxologico.it)
Enrico Molinari (enrico.molinari@unicatt.it)

Version: 4 Date: 8 May 2008

Author's response to reviews:

We formatted the paper according to the changes requested.

In order to give the right space to the large table containing studies characteristics, we changed the orientation of the corresponding pages (horizontal orientation).

Sincerely,

Dr Gian Mauro Manzoni